Author's response to reviews

Title: No association between the PREP gene and lithium responsive bipolar disorder

Authors:

Firoza Mamdani (firoza.mamdani@mail.mcgill.ca)
Adolfo Sequeira (seiado@douglas.mcgill.ca)
Martin Alda (martin.alda@mcgill.ca)
Paul Grof (pgrof@rohcg.on.ca)
Guy Rouleau (guy.rouleau@mcgill.ca)
Gustavo Turecki (gustavo.turecki@mcgill.ca)

Version: 4 Date: 26 February 2007

Author’s response to reviews:

The following amendments have been made to the manuscript, as requested by the Editors:
- The Introduction was renamed to 'Background'
- The Materials & Methods section was renamed 'Methods'
- The Competition of Interest section was renamed 'Competing interests'
- An Acknowledgments Section was added
- A figures section after the References titled 'Figure Legends' was added